

## A Clinical Severity Scoring System for Visceral Leishmaniasis in Immunocompetent Patients in South Sudan

Authors	Kämink, SS; Collin, SM; Harrison, T; Gatluak, F; Mullahzada, AW; Ritmeijer, K
Citation	A Clinical Severity Scoring System for Visceral Leishmaniasis in Immunocompetent Patients in South Sudan. 2017, 11 (10):e0005921 PLoS Negl Trop Dis
DOI	10.1371/journal.pntd.0005921
Publisher	Public Library of Science
Journal	PLoS Neglected Tropical Diseases
Rights	Archived with thanks to PLoS Neglected Tropical Diseases
Download date	03/10/2021 16:56:06
Link to Item	http://hdl.handle.net/10144/619031



## 

**Citation:** Kämink SS, Collin SM, Harrison T, Gatluak F, Mullahzada AW, Ritmeijer K (2017) A clinical severity scoring system for visceral leishmaniasis in immunocompetent patients in South Sudan. PLoS Negl Trop Dis 11(10): e0005921. https://doi.org/10.1371/journal. pntd.0005921

Editor: Diana N. J. Lockwood, London School of Hygiene and Tropical Medicine, UNITED KINGDOM

Received: February 28, 2017

Accepted: August 30, 2017

Published: October 2, 2017

**Copyright:** © 2017 Kämink et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Médecins Sans Frontières Operational Centre Amsterdam: Data are available on request to bona fide researchers via the Médecins Sans Frontières Operational Centre Amsterdam (MSF-OCA). Data were collected routinely from patients by MSF, and approval for access to these data can be granted on a case-bycase basis by the MSFOCA Ethical Review Board (ERB). This process is designed to protect patient confidentiality, which might be compromised if data were publically available. Requests to access RESEARCH ARTICLE

# A clinical severity scoring system for visceral leishmaniasis in immunocompetent patients in South Sudan

# Suzette S. Kämink<sup>1,2</sup>\*, Simon M. Collin<sup>3</sup>, Tim Harrison<sup>4</sup>, Francis Gatluak<sup>4</sup>, Abdul Wasay Mullahzada<sup>4</sup>, Koert Ritmeijer<sup>1</sup>

1 Public Health Department, Médecins Sans Frontières, Amsterdam, The Netherlands, 2 Department Health Sciences, Vrije Universiteit, Amsterdam, The Netherlands, 3 Population Health Sciences, Bristol Medical School University of Bristol, Bristol, United Kingdom, 4 Médecins Sans Frontières, Lankien, South Sudan

\* s.s.kamink@gmail.com

## Abstract

## Background

South Sudan is one of the most endemic countries for visceral leishmaniasis (VL), and is frequently affected by large epidemics. In resource-limited settings, clinicians require a simple clinical tool to identify VL patients who are at increased risk of dying, and who need specialised treatment with liposomal amphotericin B and other supportive care. The aim of this study was to develop and validate a clinical severity scoring system based on risk factors for death in VL patients in South Sudan.

## Methods

A retrospective analysis was conducted of data from a cohort of 6,633 VL patients who were treated in the Médecins Sans Frontières (MSF) hospital in Lankien between July 2013 and June 2015. Risk factors for death during treatment were identified using multivariable logistic regression models, and the regression coefficients were used to develop a severity scoring system. Sensitivity and specificity of score cut-offs were assessed by receiver operating characteristic (ROC) analysis.

## Results

In multivariable models, risk factors for death in adult VL patients were: anaemia (odds ratio (OR) 4.46 (95% CI 1.58–12.6) for Hb <6g/dL compared with  $\geq$ 9g/dL), nutritional status (OR 4.84 (2.09–11.2) for BMI <13 kg/m<sup>2</sup> compared with  $\geq$ 16 kg/m<sup>2</sup>), weakness (OR 4.20 (1.82–9.73) for collapsed compared with normal weakness), jaundice (OR 3.41 (1.17–9.95)), and oedema/ascites (OR 4.86 (1.67–14.1)). For children and adolescents the risk factors were: age (OR 10.7 (6.3–18.3) for age <2 years compared with 6–18 years), anaemia (OR 7.76 (4.15–14.5) for Hb <6g/dL compared with  $\geq$ 9g/dL), weakness (OR 3.13 (22.8–105.2) for collapsed compared with normal weakness), and jaundice (OR 12.8 (4.06–40.2)). Severity scoring predictive ability was 74.4% in adults and 83.4% in children and adolescents.

data should be made to the Medical Director of MSF-OCA by mentioning the study code 1651: Dr. Sid Wong, Medical Director, Médecins Sans Frontières, Operational Centre Amsterdam Plantage Middenlaan 14, 1018 DD Amsterdam. Email: sidney.wong@amsterdam.msf.org

**Funding:** The authors received no specific funding for this work.

**Competing interests:** The authors have declared that no competing interests exist.

## Conclusion

Our evidenced-based severity scoring system demonstrated sufficient predictive ability to be operationalised as a clinical tool for rational allocation of treatment to VL patients at MSF centres in South Sudan.

#### Author summary

Visceral leishmaniasis (VL), also known as kala-azar, is a neglected tropical disease caused by a parasite (*Leishmania donovani*) and transmitted to humans through the bite of a sandfly. South Sudan is one of the highest endemic countries for VL, frequently affected by epidemics. In South Sudan are different treatment options for VL available: the standard therapy given on ambulatory bases by intramuscular injections for 17 days, and specialized treatment for patients with severe VL, administered intravenously in a hospital over 12 days. In the extremely resource limited context of South Sudan, the most optimal treatment to patients with severe VL should be provided, but a rational use of drugs maintained. In this study, we identified risk factors for death in VL patients of South Sudan, and based on these risk factors we developed a severity scoring system. This severity scoring system will be a clinical decision making tool for allocation of VL patients to the appropriate treatment and to minimise the mortality of the VL patients in South Sudan.

### Introduction

Visceral leishmaniasis (VL, or kala-azar) is a neglected tropical disease which is caused by the obligate intracellular protozoa of the *Leishmania donovani* complex (*L. donovani* and *L. infan-tum*) (1). In East Africa VL is caused by *L. donovani*, and is transmitted anthroponotically by phlebotomine sandfly vectors [1].

VL targets the lymphatic and reticuloendothelial system, affecting spleen, liver, mucosa of the small intestine and respiratory tract, bone marrow, lymph nodes and the other lymphoid tissues, causing persistent fever, organomegaly, pancytopenia and wasting. Patients with VL are severely immunocompromised, and death occurs from opportunistic or concomitant infections, or from complications such as malnutrition, anaemia or bleeding. Without treatment VL is typically fatal [1, 2].

South Sudan is one of the most endemic countries for VL, with an annual incidence of 7,400–14,200 cases [3]. The country has been affected by frequent large epidemics, often associated with mass displacement due to armed conflict, and causing high mortality [4, 5].

Current first line treatment for VL in South Sudan is the pentavalent antimonial, sodium stibogluconate (SSG), in combination with an aminoglycoside, paromomycin (PM). SSG/PM is given on an ambulatory basis over 17 days with daily intramuscular injections [6, 7]. However, SSG is often poorly tolerated, and toxicity results in a significant incidence of serious adverse events such as pancreatic, hepato- and nephrotoxicity, cardiotoxicity, gastro-intestinal disorders and, in pregnant women, spontaneous abortion [8–11]. For this reason, SSG is contraindicated for specific patient groups (e.g. pregnant women or HIV co-infected) or in patients with severe VL. These patients should be treated with liposomal amphotericin B (AmBisome) [12]. AmBisome is much better tolerated but is much more expensive and requires cold chain transportation, cool storage, intravenous administration, and hospitalisation for at least 12 days.

These are major challenges in the resource limited context of South Sudan, meaning that rational use of AmBisome is currently a key operational requirement. AmBisome needs to be reserved for severely ill VL patients who are at high risk of dying or at risk of SSG intolerability [12–14], whilst standard SSG/PM treatment can continue to be administered to patients with uncomplicated VL [7]. In resource-limited settings, clinicians require a simple clinical tool to identify VL patients who are at increased risk of dying, and who need specialised treatment with liposomal amphotericin B and other supportive care.

The aim of this study was to develop an evidenced-based risk scoring system which could be used as a clinical decision making tool in the field, to help clinicians decide whether a VL patient requires intensive VL care and treatment with AmBisome or standard VL management and less intensive monitoring. The risk scoring system would be based on risk factors for death during treatment in a retrospective cohort of VL patients, and would be validated internally and against older cohorts of VL patients treated by MSF in South Sudan.

## Methods

## Ethics statement

This research fulfilled the exemption criteria set by the Médecins Sans Frontières Ethical Review Board for a posteriori analyses of routinely collected clinical data, and thus did not require MSF ERB review. It was conducted with permission from the Medical Director of the MSF Operational Centre Amsterdam. All data were anonymised before analysis.

## Study design and data source

A retrospective analysis was conducted of routinely collected data from a cohort of 6,821 VL patients who attended the hospital of Médecins Sans Frontières (MSF) in Lankien, Jonglei state, South Sudan during the VL outbreak between July 2013 and June 2015. The data were stored in an electronic database, and included key dates, demographic, anthropometric, diagnostic and clinical characteristics of patients, treatment regime and outcome. Patients with incomplete data, 8, and who defaulted from treatment were excluded from analyses. VL-HIV coinfected patients were excluded because treatments and outcomes are different in this small immunocompromised subgroup (during the study period the HIV/VL rate in Lankien was only 0.15%).

### Outcome

Our analysis was based on a binary outcome of died (during or immediately after treatment) or survived treatment. "Survived" means that the patient was discharged after successful clinical response to treatment: absence of fever, reduction of spleen and liver size, increased haemoglobin, restored appetite, and feeling well. Patients with an increased risk of treatment failure or relapse (i.e. patients with a prior episode of VL, or patients with inadequate or doubtful clinical response) require a negative parasitological test-of-cure by aspirate microscopy to confirm cure of VL.

There is no systematic follow up of patients after discharge, because they return to areas remote from the treatment centre. A defaulter was defined as a patient who did not complete treatment, and had an unknown outcome.

### **Risk factors**

The dataset included the following potential risk factors for death in VL: age (years), sex, presence/absence of jaundice, lymphadenopathy, oedema/ascites; prior episode/relapse of VL, Hb level (g/dL), spleen size (cm below the left costal margin), self-reported duration of illness (months), and nutritional status (body mass index (BMI) in patients  $\geq$ 19years (kg/m<sup>2</sup>) and weight for height Z scores (WHZ) in patients <19 years old). State of weakness, was defined as 'normal weakness' (non-severe); 'severe weakness' (if patient needs assistance in walking or, in babies, if unable to sit up); or 'collapse' (if patient is unable to sit or drink or, in babies, if hypotonic and unable drink unaided). All of these variables have been identified as risk factors for death in earlier VL patient cohorts from South Sudan [2, 5, 15, 16].

## Diagnosis

All patients presenting with a history of fever more than 2 weeks and splenomegaly and/or lymphadenopathy and/or wasting (BMI <16 mg/m2 or <-2 Zscore) were considered VL suspects and underwent further diagnostic evaluations. Patients without prior VL treatment history (suspect primary VL) were first screened using the rK39 rapid diagnostic test (IT-Leish, Bio-Rad laboratories, USA) and a positive result confirmed VL. Those testing negative were screened with the leishmania direct agglutination test (DAT, Royal Tropical Institute, Amsterdam, The Netherlands) and a high titer ( $\geq$ 1:6400) confirmed VL. Those with an intermediate DAT titer (1:800–1:3200) underwent tissue aspiration (spleen or lymph node) and positive result confirmed VL. Patients with prior VL treatment history (suspect relapse VL) underwent tissue aspiration and a positive result confirmed VL. A clinical diagnosis was made in patients contra-indicated for spleen aspirate (i.e. severely anaemic, bleeding, pregnant or collapsed) and didn't have palpable lymph nodes or those with negative lymph node aspirate results but with persistent strong VL clinical suspicion in the absence of differential diagnoses [17].

## Treatment

The first line treatment was with SSG (20 mg/kg) in combination with PM (15 mg/kg) given on an ambulatory basis over 17 days with daily intramuscular injections.

Specific patients groups with contra-indication for SSG (pregnant women, HIV coinfected) or patients with known poor tolerability of SSG (age >45 years, severe VL) were treated with AmBisome by 6 IV infusions of 5 mg/kg on alternate days. Treatment had started without delay on the same day as confirmation of diagnosis.

If clinical and laboratory investigations confirm severe VL disease, specialised treatment was started immediately. 'Specialised treatment' means besides AmBisome also rehydration, aggressive antibiotic treatment (ceftriaxone IV), other supportive treatment (e.g. for malnutrition), intensified monitoring and treatment of any suspected abnormality.

As the aim is to predict a patient's prognosis at the time of diagnosis, the treatment was not included in the analysis.

## Statistical analysis

**Risk factors.** Univariate and multivariable logistic regression models were used to quantify the associations of risk factors with death as odds ratios (ORs) with 95% confidence intervals (CIs). Two prediction models were built: one with patients aged 19 years and older (adults), and one with the patients younger than 19 years (defined as children and adolescents). Variables were included in the multivariable analysis if they had a significance level of p < 0.2 in the univariate analysis, and backwards elimination was then used to build final prediction models.

**Derivation of the severity scoring system.** The severity scoring system was developed using a method described by Barquet et al, which is based on the Spiegelhalter-Knills Jones method [18–20]. Regression coefficients (RCs) from the final multivariable prediction model

were used to calculate a score for each level of each risk factor (by dividing each RC by the smallest RC in the model and rounding up or down to the nearest integer). The reference level of each categorical variable was scored as '0'. The severity score for an individual patient is the sum of his/her severity scores for each risk factor; a higher score indicating higher risk of death, i.e. more severe illness [18, 21]. Probabilities of death can be calculated for each severity score, but the rounding up or down of each RC means that combinations of different levels of risk factors can lead to the same severity score, i.e. for each severity score there is a range of probabilities of death. The upper and lower limits of this range for each severity score was calculated using the method described by Coura-Vital et al [22].

**Discriminative ability of the risk scoring system.** The discriminative (predictive) ability of the severity scoring system was assessed using receiver operating characteristic (ROC) analysis [25, 26]. Discriminative ability was quantified by area under the curve (AUC), categorised as: 'no better than a random guess' (AUC 0.5–0.6); 'poor' (AUC 0.6–0.7); 'fair' (AUC 0.7–0.8); 'good' (AUC 0.8–0.9) or 'excellent' (AUC 0.9–1.0) [23, 24].

**Determination of severity score thresholds.** ROC analysis was used to calculate the sensitivity and specificity of different risk score thresholds [25, 26]. The optimum threshold can be determined by considering the clinical and operational implications of the sensitivity and specificity of different thresholds.

**Validation.** The severity scoring system was validated in three datasets from previous studies in South Sudan: two from patients treated in Lankien during 1999–2002 (N = 708) and during 2002–2005 (N = 1882), and one from patients treated in Malakal during 2002–2005 (N = 1757) [2, 15]. Validation was performed with the use of the Z test by comparing the predictive ability (AUC) of the severity scoring system across the datasets [27].

**Statistical software.** SPSS for Windows 23.0 and STATA/IC 14.0 were used for the statistical analyses.

## Results

#### Study population

The initial dataset included 6,821 VL patients. After excluding patients with incomplete data (n = 8), HIV co-infection (n = 11), who defaulted from treatment (n = 159) or referred to another non MSF facility (n = 10) the total sample size was 6,633. Of these, 3,631 (54.7%) were male and 3,002 (45.3%) were female. Out of the 6,614 patients of whom the duration of illness was known 6,087 (92%) presented within 1 month after onset of symptoms, and no patient presented later than 6 months.

Of the 6,615 patients whose treatment regime was known, 5,149 patients were treated with SSG/PM and 1,466 patients with AmBisome. The data comprised 4,931 (74.3%) children and adolescents (aged < 19 years) and 1,702 (25.7%) adults ( $\geq$ 19 years). Mortality data was captured during the complete time of admission in the hospital, until discharge or death; the longest admission duration was 134 days. In total 6,447 patients (97.7%) survived and 186 (2.8%) died during treatment; 33% (49/186) of the deaths occurred within 48 hours of admission.

The characteristics of patients in each age group who died compared with those who survived are shown in Table 1. Mortality in children and adolescents was 2.4% compared with 4.1% among adults (OR 1.78 (95% CI 1.32–2.41)).

## Risk factors in adults

Univariate analysis showed that age, Hb, state of weakness, nutritional status, jaundice, and oedema/ascites were strongly associated with VL mortality (Table 2): patients >45 years old had 3.4-fold higher odds of death (OR 3.42 (95% CI 1.64–7.16)) compared with patients aged

		Adults (patients >	≥19 years old)	Ch Children and Adolescents (patients <19 years old)			
Variable		Died n = 70 (4.1%)	Survived n = 1632 (95.9%)		Died n = 116 (2.4%)	Survived n = 4815 (97.5%)	
Age, years							
	18–25	11 (2.2)	493 (97.8)	6–18	22 (0.7)	2949 (99.2)	
	26–35	20 (3.2)	602 (96.8)	2–5	37 (2.5)	1424 (97.5)	
	36–45	17 (6.4)	247 (93.6)	<2	57 (11.4)	442 (88.6)	
	> 45	22 (7.1)	290 (92.9)				
Nutrition status	BMI, kg/m <sup>2</sup>			WHZ			
	≥ <b>16</b>	14 (1.7)	806 (98.3)	>-2	27 (2.3)	1137 (97.7)	
	14.5–15.9	25 (5.1)	468 (94.9)	≤-2	35 (2.1)	1619 (97.9)	
	13–14.4	18 (6.5)	259 (93.5)	≤-3	32 (1.3)	1652 (98.7)	
	< 13	13 (11.6)	99 (88.4)	≤-4	21 (6.1)	323 (93.9)	
Sex							
	male	42 (4.4)	915 (95.6)	male	59 (2.2)	2615 (97.8)	
	female	28 (3.9)	717 (96.1)	female	57 (2.5)	2200 (97.5)	
State of weakness							
	normal weakness	36 (2.6)	1345 (97.4)	normal weakness	83 (1.8)	4563 (98.2)	
	severe weakness	24 (9.1)	239 (90.9)	severe weakness	28 (10.5)	239 (89.5)	
	collapse	10 (17.2)	48 (82.8)	collapse	5 (41.7)	7 (58.3)	
Hb, g/DI							
	≥ <b>9</b>	37 (3.1)	1174 (96.9)	≥ <b>9</b>	25 (1.0)	2562 (99.0)	
	7.5–8.9	17 (5.2)	310 (94.8)	7.5–8.9	29 (2.1)	1352 (97.9)	
	6–7.4	10 (7.4)	126 (92.6)	6–7.4	34 (4.3)	749 (95.7)	
	< 6	6(21.4)	22 (79.6)	< 6	28 (15.6)	151 (84.3)	
Duration of illness, months							
	= 1	49 (3.9)	1195 (96.1)	= 1	88 (2.3)	3714 (97.7)	
	< 1	12 (4.3)	268 (95.7)	<1	24 (3.2)	737 (96.8)	
	≥2	9 (5.2)	164 (94.8)	≥ <b>2</b>	4 (1.1)	350 (98.9)	
Jaundice	no	64 (3.8)	1613 (96.2)	no	111 (2.3)	4790 (97.7)	
	yes	6 (24.0)	19 (76.0)	yes	5 (17.9)	23 (82.1)	
Oedema/ascites	no	64 (3.8)	1610 (96.2)	no	110 (2.3)	4766 (97.8)	
	yes	6 (21.4)	22 (79.6)	yes	6 (10.9)	49 (89.1)	
Lymphadenopathy	no	20 (5.6)	337 (94.4)	no	37 (4.0)	898 (96.0)	
	yes	50 (3.7)	1290 (96.3)	yes	79 (2.0)	3907 (98.0)	
Prior episode VL	0	61 (4.0)	1475 (96.0)	0	99 (2.2)	4347 (97.8)	
	1 or more	9 (5.4)	157 (94.6)	1 or more	17 (3.5)	468 (96.5)	
Spleen size, cm							
	< 1	31 (4.9)	596 (95.1)	< 1	40 (2.7)	1459 (97.3)	
	1-4	21 (4.0)	502 (96.0)	1-4	50 (2.5)	1936 (97.5)	
	5–7	10 (3.7)	262 (96.3)	5–7	15 (1.8)	803 (98.2)	
	>7	8 (2.9)	268 (97.1)	>7	11 (1.8)	609 (98.2)	

#### Table 1. Characteristics of the study population by outcome (survived or died).

https://doi.org/10.1371/journal.pntd.0005921.t001

18–25 years; patients in a state of collapse were 8 times more likely to die (OR 7.80 (95% CI 3.66–16.6)) compared with patients who arrived in a 'normal' state of weakness; Hb levels < 6g/dL increased the odds of dying almost 9-fold (OR 8.67 (95% CI 3.32–22.6)) compared with

levels  $\geq$ 9g/dL; and BMI <13 kg/m<sup>2</sup> was associated with 7.6-fold higher odds of death (OR 7.56 (95% CI 3.45–16.6)) compared with BMI  $\geq$  16 kg/m<sup>2</sup>. Presence of jaundice and oedema/ascites increased the odds of dying 8-fold (OR 7.95 (95% CI 3.07–20.6)) and almost 7-fold (OR 6.87 (95% CI 2.69–17.5)), respectively. Sex, lymphadenopathy, prior episode VL, duration of illness, and spleen size were not associated with risk of death.

The effects of Hb, state of weakness, nutritional status, jaundice, and oedema/ascites were reduced by mutual adjustment, and age was eliminated from the final prediction model (Table 2). In this model, patients in a state of collapse were 4 times more likely to die (OR 4.20 (95% CI 1.82–9.73)) compared with patients who arrived in a 'normal' state of weakness; Hb

Variable (n = 1,702)	Category	Univariate logistic regression			Final prediction model		
		Crude OR <sup>a</sup> (95% Cl <sup>b</sup> )	P-value <sup>c</sup>	RC <sup>d</sup>	Adjusted OR (95%CI)	P-value <sup>c</sup>	
Sex	male		0.52				
	female	0.85(0.52-1.39)					
Age, years	18–25		0.002				
	26–35	1.50 (0.71–3.15)					
	36–45	3.10 (1.43–6.73)					
	> 45	3.42 (1.64–7.16)					
State of weakness	normal weakness		< 0.001				
	severe weakness	3.76 (2.20–6.41)		0.90	2.45 (1.39–4.33)	0.002	
	collapse	7.80 (3.66–16.6)		1.44	4.20 (1.82–9.73)	0.001	
BMI, kg/m <sup>2</sup>	≥ <b>16</b>		< 0.001				
	14.5–15.9	3.08 (1.58–5.98)		0.96	2.60 (1.32-5.14)	0.006	
	13–14.4	4.00 (1.96-8.16)		1.02	2.78 (1.31–5.91)	0.008	
	< 13	7.56 (3.45–16.6)		1.58	4.84 (2.09–11.2)	< 0.001	
Hb, g/dL	≥9		< 0.001				
	7.5–8.9	1.74 (0.97–3.14)		0.07	1.08 (0.57–2.03)	0.82	
	6–7.4	2.52 (1.23–5.19)		0.42	1.52 (0.70–3.30)	0.29	
	< 6	8.67 (3.32–22.6)		1.50	4.46 (1.58–12.6)	0.005	
Oedema/ascites	no		< 0.001			0.004	
	yes	6.87 (2.69–17.5)		1.58	4.86 (1.67–14.1)		
Jaundice	no		< 0.001			0.025	
	yes	7.95(3.07–20.6)		1.23	3.41 (1.17–9.95)		
Lymphadenopathy	no		0.12				
	yes	0.65(0.38-1.11)					
Prior episode VL	0		0.37				
	1 or more	1.39 (0.68–2.85)					
Duration of illness, months	= 1		0.73				
	<1	1.09 (0.57–2.08)					
	≥2	1.34 (0.65–2.78)					
Spleen size, cm	<1		0.53				
	1–4	0.80 (0.46–1.42)					
	5–7	0.73 (0.35–1.52)					
	>7	0.57 (0.26–1.27)					

#### Table 2. Risk factors for death among adult patients (>19 years old).

<sup>a</sup> OR indicates odds ratio

<sup>b</sup> CI indicates confidence interval

<sup>c</sup> P value from Chi squared test

<sup>d</sup> RC indicates (unexponentiated) regression coefficient

levels <6g/dL increased the odds of dying 4.5-fold (OR 4.46 (95% CI 1.58–12.6)) compared with levels  $\ge 9g/dL$ ; and BMI <13 kg/m<sup>2</sup> was associated with almost 5-fold higher odds of death (OR 4.84 (95% CI 2.09–11.2)) compared with BMI  $\ge 16$  kg/m<sup>2</sup>. Presence of jaundice and oedema/ascites increased the odds of dying >3-fold (OR 3.41 (95% CI 1.17–9.95)) and almost 5-fold (OR 4.86 (95% CI 1.67–14.1)), respectively.

## Risk factors in children and adolescent

The univariate analyses of children and adolescents showed that age, Hb, state of weakness, jaundice, oedema/ascites and WHZ were strongly associated with VL mortality (Table 3).

#### Table 3. Risk factors for death among child and adolescent patients (<19 years old).</th>

		Univariate logistic	regression		Final prediction model	
Variable (n = 4,931)	Category	Crude OR <sup>a</sup> (95% Cl <sup>b</sup> )	P <sup>c</sup> value	RCd	Adju <mark>ste</mark> d OR (95%CI)	P value <sup>c</sup>
Sex	male		0.46			
	female	1.15 (0.79–1.66)				
Age, years	6–18		< 0.001			
	2–5	3.48 (2.05–5.93)		0.98	2.67 (1.55-4.61)	< 0.001
	<2	17.29 (10.46–28.6)		2.37	10.74 (6.31–18.3)	< 0.001
State of weakness	normal weakness		< 0.001			
	severe weakness	6.44 (4.12–10.1)		0.92	2.53 (1.53–4.18)	< 0.001
	state of collapse	39.27(12.21-126.3)		3.13	22.82 (4.95–105.2)	< 0.001
WHZ	> -2		< 0.001			
	= -2	0.91 (0.55–1.51)				
	= -3	0.87 (0.49–1.37)				
	= -4	2.74 (1.58–4.90)				
Hb, g/dL	≥9		< 0.001			
	7.5–8.9	2.20 (1.28–3.77)		0.57	1.77 (1.02–3.08)	0.044
	6–7.4	4.65 (2.76–7.85)		1.02	2.77 (1.59–4.82)	< 0.001
	< 6	19.00 (10.81–33.4)		2.05	7.76 (4.15–14.5)	< 0.001
Oedema/ascites	no		<0.001			
	yes	5.31 (2.25–12.7)				
Jaundice	no		< 0.001			< 0.001
	yes	9.38 (3.50–25.2)		2.55	12.77 (4.06–40.2)	
Lymphadenopathy	no		< 0.001			
	yes	0.49 (0.33–0.73)				
Prior episode VL	0		0.08			
	1 or more	1.60 (0.94–2.69)				
Duration of illness, months	= 1		0.12			
	< 1	1.37 (0.87–2.17)				
	≥2	0.48 (0.18–1.32)				
Spleen size, cm	< 1		0.44			
	1–4	0.94 (0.17–1.44)				
	5–7	0.68 (0.37–1.24)				
	>7	0.66 (0.34–1.29)				

<sup>a</sup> OR indicates odds ratio

<sup>b</sup> CI indicates confidence interval

<sup>c</sup> P value from Chi squared test

<sup>d</sup> RC indicates (unexponentiated) regression coefficient

Patients < 2 years old had >17-fold higher odds of death (OR 17.3 (95% CI 10.5–28.6)) compared with patients aged 6–18 years; patients in a state of collapse were 39 times more likely to die (OR 39.3 (95% CI 12.2–126.3)) compared with patients who arrived in a 'normal' state of weakness; Hb levels <6g/dL increased the odds of dying 19-fold (OR 19.0 (95% CI 10.8–33.4)) compared with levels  $\geq$ 9g/dL; and WHZ <-4 was associated with almost 3-fold higher odds of death (OR 2.74 (95% CI 1.58–4.90)) compared with WHZ > -2. Presence of jaundice and oedema/ascites increased the odds of dying >9-fold (OR 9.38 (95% CI 3.50–25.2)) and >5-fold (OR 5.31 (95% CI 2.25–12.7)), respectively.

For patients with lymphadenopathy, the odds of dying reduced by half. (OR 0.49 (95% CI 0.33–0.73) Sex, prior episode VL, and spleen size were not associated with risk of death. Duration of illness did not have sufficient cases for analyses.

The effects of age, Hb, state of weakness and jaundice were reduced by mutual adjustment, and lymphadenopathy, nutritional status and oedema/ascites were eliminated from the final prediction model (Table 3). In this model, patients < 2 years old had almost 11-fold higher odds of death (OR 10.7 (95% CI 6.31–18.26)) compared with patients aged 6–18 years; patients in a state of collapse were almost 23 times more likely to die (OR 22.8 (95% CI 4.95–105.2)) compared with patients who arrived in a 'normal' state of weakness; Hb levels <6g/dL increased the odds of dying nearly 8-fold (OR 7.76 (95% CI 4.15–14.5)) compared with levels  $\geq$  9g/dL; and presence of jaundice increased the odds of dying 3.4-fold (OR 3.41 (95% CI 1.17–9.95)).

## Severity scoring system

The scores for each variable in the final prediction model are shown in Table 4, and the range of probabilities for each score (due to rounding up or down of regression coefficients) are presented in Table 5. For adults, probability of death exceeded 10% for risk scores  $\geq$ 3 (6.1% (104/ 1695) of adults); in children and adolescents, the threshold for exceeding a 10% probability of death was a score  $\geq$ 6 (6.8% (339/4921) of children).

			Severity Score
Variable	Category	Adults	Children and adolescents
State of weakness	normal weakness	0	0
	severe weakness	1	2
	state of collapse	2	5
Hb, g/dL	≥ 9.0	0	0
	7.5–8.9	0	1
	6.0–7.4	0	2
	< 6.0	2	4
BMI, kg/m <sup>2</sup>	≥ 16	0	
	14.5–15.9	1	
	13.0–14.4	1	
	< 13	2	
Jaundice	not present	0	0
	present	1	4
Oedema/ascites	not present	0	
	present	2	
Age, years	6–18		0
	2–5		2
	<2		4

Table 4. Severity scoring for adults ( $\geq$ 19 years), and children	en and adolescents (< 19 years).
---	----------------------------------

	Adults			Children and adolescents				
			Probability	of death			Probability	of death
Severity score	All cases (n)	Observed case fatality (n)	Lower limit	Upper limit	All cases (n)	Observed case fatality (n)	Lower limit	Upper limit
0	719	10	1.2%	1.4%	1712	8	0.4%	0.5%
1	640	20	2.9%	4.0%	745	6	0.7%	0.7%
2	232	19	4.9%	10.3%	1001	9	1.0%	1.1%
3	77	13	11.8%	22.1%	450	6	1.8%	2.0%
4	21	6	18.6%	35.9%	506	13	2.7%	5.0%
5	6	2	37.3%	57.9%	168	17	4.8%	8.6%
6	-	-	52.5%	71.4%	219	26	7.2%	14.3%
7–8	-	-	74.2%	89.5%	95	20	17.8%	36.1%
9–10	-	-	94.8%	94.8%	19	7	37.5%	47.1%
11-17	-	-	-	-	6	4	46.1%	99.0%

#### Table 5. Probabilities of death<sup>a</sup> at each score for adults ( $\geq$ 19 years), and children and adolescents (< 19 years)<sup>b</sup>.

<sup>a</sup>Low risk (green)[19] <3% probability of death; Moderate risk (yellow) 3–10% probability of death; High risk (red) >10% probability of death. <sup>b</sup> The number of cases in this table (adults N = 1695, children/adolescents N = 4921) are slightly lower than shown in Table 1 (adults N = 1702, children/ adolescents N = 4931) due to missing data for some of the predictive variables.

https://doi.org/10.1371/journal.pntd.0005921.t005

For the severity scoring of the adults the classification matrix resulted in 80.8% correctly predicted deaths. The AUC of the severity scores of the adults gave an overall predictive performance of 74.4% (95 CI 68.0%-81.0%), indicating 'fair' predictive ability (Fig 1). For children and adolescents, the classification matrix showed 80.6% correctly predicted. The AUC was 83.4% (95% CI 78.0%-86.8), interpreted as 'good' predictive ability (Fig 2). Sensitivity >55% required a score  $\geq 2$  in adults (sensitivity 57%, specificity 82%) and  $\geq 5$  in children and adolescents (sensitivity 64%, specificity 91%). Sensitivity >75% required a score  $\geq 1$  in adults (sensitivity 86%, specificity 44%) and  $\geq 4$  in children and adolescents (sensitivity 75%, specificity 81%) (Table 6).

![](_page_10_Figure_7.jpeg)

![](_page_10_Figure_8.jpeg)

![](_page_11_Figure_2.jpeg)

Fig 2. Area under the curve for children and adolescents scoring.

https://doi.org/10.1371/journal.pntd.0005921.g002

### Validation

External validation of the scoring for adults with the Lankien datasets of 1999–2002, 2002–2005 and the Malakal dataset of 2002–2005 yielded AUC of 72.2%, 79.5% and 71.2%, respectively. Discriminative ability did not differ significantly across the four datasets (p = 0.48). For children and adolescents, corresponding AUC were 72.2%, 82.8% and 76.6%, with only very weak evidence of a difference in discriminative ability across the three datasets (p = 0.13).

## Discussion

In this study, the risk factors for death in VL patients in South Sudan during the VL epidemic from 2013 to 2015 were analysed. Significant risk factors for adult patients were nutritional status (BMI), Hb, weakness status, jaundice and oedema/ascites. In children and adolescents, the

Table 6. Performance of the severity scoring for prediction of death in VL patients in South Sudan.

Severity score	Ad	ults	Children and	adolescents
	Sensitivity	Specificity	Sensitivity	Specificity
0	100%	0.0%	100%	0.0%
1	85.7%	43.7%	-	-
2	57.1%	81.8%	87.9%	50.8%
3	30.0%	94.9%	80.2%	71.5%
4	11.4%	98.8%	75.0%	80.7%
5	2.9%	99.8%	63.8%	91.0%
6	0.0%	100%	49.1%	94.1%
7	0.0%	100%	26.7%	98.1%
8	0.0%	100%	25.0%	98.5%
9–10	0.0%	100%	9.5%	99.8%
11	-	-	3.4%	100%
12	-	-	2.6%	100%
13	-	-	1.7%	100%
14	-	-	0.0%	100%

risk factors were age, Hb, weakness status and jaundice. Using these risk factors, an evidencebased clinical severity scoring system was developed that may be able to determine reliably and easily a patient's risk of dying, thereby enhancing the rational use of more costly and complex VL treatments. The overall accurate predictive ability of this severity scoring system was confirmed by external validation in other data from the same setting.

VL treatment effect was not analysed, because in observational studies indications for treatment are usually not standardised, and confounding by indication could lead to bias. Moreover, as the aim is to predict a patient's prognosis at the time of diagnosis, the treatment should not be included.

In operationalising a risk scoring system, erring towards a higher sensitivity over a lower specificity would mean that >28% of child and > 56% of adult patients should receive specialised VL care. This is unfortunately not possible in a severely resource limited setting such as South Sudan. The optimal threshold will be a compromise between sensitivity and specificity, i.e. a threshold needs to be chosen that includes as many patients with increased risk of dying as possible, whilst maintaining a rational use of scarce resources. Whilst our focus has been to reduce VL mortality by identifying those patients most at risk of dying, the severity scoring system could be a useful tool in the management of patients at lower risk. VL patients with a low risk of dying (<3.0%), could be treated on an ambulatory basis in outpatient treatment centres. Patients with a moderate risk of dying (3–10%) could be treated in an outpatient department, but under close monitoring by experienced clinicians. Patients with a higher score and thus a higher risk of dying (>10%) should be admitted as inpatients for specialised VL care and intensive monitoring.

In this study there was no association of acute malnutrition (WHZ <-2) with death in children and adolescents, although severe acute malnutrition (WHZ <-4) was associated with increased risk of death. This finding is in contrast with earlier studies in South Sudan in 2004 and 1991 [15, 16]. The lack of association at less severe levels of malnutrition in our study could be explained by improved treatment of malnourished children in an intensive therapeutic feeding centre (ITFC), where malnourished children receive specialised medical and nutritional care. This specialized treatment may have mediated the impact of malnutrition on VL mortality in children, except for the very severely malnourished. In our final prediction model, severely malnourished children would score highly for being in a state of collapse and/or presenting with very low Hb levels, and would exceed the threshold for specialised care.

A finding which is in line with previous studies in South Sudan is that young age was a strong predictor for VL mortality [15, 16]. The eleven-fold higher odds of death in children below 2 years (compared with older children) demonstrate the vulnerability of this youngest age group. Accordingly, age <2 years contributed 4 points, and even without additional risk factors already is at 'moderate risk'. On the other hand, older age (>45 years), which was strongly associated with mortality in the univariate analysis, was not retained as an independent risk factor in the multivariable regression model, contrary to earlier studies [15, 16, 22]. In these earlier studies, most elderly patients were treated with SSG, whereas in the present study 87% of the patients older than 45 years were treated with AmBisome [15, 16, 28]. Several studies had demonstrated a high mortality in older patients due to SSG toxicity, and during those earlier studies there was no or only limited AmBisome available [12]. Therefore in our dataset, AmBisome may have mediated the effect of older age on VL mortality. This seems to confirm the recommendation that treatment with AmBisome may be lifesaving for the elderly VL patients [9].

In contrast with two previous studies we did not find that splenomegaly was associated with increased risk of death [15, 29]. Conversely, we found a crude inverse association of lymphade-nopathy with risk of death (although not evident when adjusted for other risk factors). We can think of no plausible explanation for these apparently anomalous findings.

The mortality rate in this study cohort was much lower than in earlier studies (2.8% vs. 10.9% and 10.0%) [7, 16]. Despite the fact that the difficult context of civil war, violence and displacement, the weak and unstable health care system was similar between this study and the earlier studies in South Sudan [30, 31]. This may be partially explained by earlier presentation at the Lankien hospital: more than 92% of the patients were able to present early, i.e. within one month after onset of clinical disease, whereas only 65% of the patient were able to during those earlier studies. As there were only 13 deaths in 527 patients presenting more than 1 month after onset of symptoms, duration of illness could not be identified as a risk factor. This seems to support the interpretation that specific joint efforts led since 2009 by the World Health Organisation in collaboration with Ministry of Health and nongovernmental organisations (NGOs), have been successful in achieving improved access to VL care in South Sudan (by decentralising VL treatment services and ensuring supply of tests and drugs) [3, 30].

The main strength of our study is that it was based on a large dataset with very few patients excluded because of missing data. Also, a robust external validation was possible, because of the availability of MSF datasets from previous VL outbreaks in South Sudan. The study was based on a cohort of South Sudanese patients, and therefore the predictors of death and the severity scoring system should not be generalised to patient populations in other countries. For example, where VL is caused by other Leishmania strains, where HIV co-infection is more prevalent, or where resistance to SSG is more common or AmBisome is more affordable. Another limitation is that this retrospective study was conducted with routinely collected programme data. Therefore it may be missing out on other important risk factors that were not included in this database, such as laboratory parameters (e.g. related to electrolyte disturbances, blood cell counts, or immune status). Although one third of deaths occurred within the first 48 hours after admission, which may limit the impact of our severity scoring system on mortality, the predictive ability and simplicity of the system means that it can be easily operationalised and implemented in the field.

The scoring system presented in this study only includes clinical parameters and a simple Hb lab test, and it therefore presents a practical tool that can be used in all field hospitals and health centres in the VL endemic areas in South Sudan. Clinicians will use their clinical judgement and experience to make treatment decisions, aided by the risk scoring tool. Given the strong associations between known risk factors and mortality, it would not be ethical to attempt a randomised trial of the severity scoring system, but we would hope to see a (continued) overall improvement in treatment outcomes in VL programmes in South Sudan until new safe, effective, and affordable treatment becomes available for all patients.

## Supporting information

**S1 File. STROBE checklist.** (DOCX)

**S1 Fig. VL patient receives the painful SSG injection in Pieri, Jonglei state, South Sudan, 2013.** Image credit: M.den Boer, MSF. (TIF)

## Acknowledgments

The authors wish to thank Maiza Campos Ponce (Vrije University Amsterdam), Ruby Siddiqui (MSF, London), Ahmed Abdi Dahir and Geke Huisman (MSF medical coordination in Juba), and all MSF staff that have contributed to care of the VL patients in Lankien hospital over the past years.

## **Author Contributions**

Conceptualization: Koert Ritmeijer.

Data curation: Tim Harrison.

Formal analysis: Suzette S. Kämink, Simon M. Collin, Koert Ritmeijer.

Funding acquisition: Koert Ritmeijer.

Investigation: Tim Harrison, Francis Gatluak, Abdul Wasay Mullahzada.

Methodology: Suzette S. Kämink, Simon M. Collin, Koert Ritmeijer.

Project administration: Abdul Wasay Mullahzada.

**Resources:** Suzette S. Kämink, Simon M. Collin, Tim Harrison, Francis Gatluak, Abdul Wasay Mullahzada, Koert Ritmeijer.

Software: Suzette S. Kämink.

Supervision: Koert Ritmeijer.

Validation: Suzette S. Kämink, Simon M. Collin, Koert Ritmeijer.

Visualization: Suzette S. Kämink, Koert Ritmeijer.

Writing - original draft: Suzette S. Kämink, Koert Ritmeijer.

Writing - review & editing: Suzette S. Kämink, Simon M. Collin, Koert Ritmeijer.

#### References

- World Health Organisation. Control of the leishmaniasis: report of the meeting of the WHO Expert Committee on the Control of Leishmaniases, WHO technical report series, no. 949. Geneva, 22–26 March 2010.
- Gorski S, Collin SM, Ritmeijer K, Keus K, Gatluak F, Mueller M, et al. Visceral leishmaniasis relapse in Southern Sudan (1999–2007): a retrospective study of risk factors and trends. PLoS Negl Trop Dis. 2010; 4(6):e705. https://doi.org/10.1371/journal.pntd.0000705 PMID: 20544032
- 3. Alvar J, Velez ID, Bern C, Herrero M, Desjeux P, Cano J, et al. Leishmaniasis worldwide and global estimates of its incidence. PloS One. 2012; 7(5):e35671. https://doi.org/10.1371/journal.pone.0035671 PMID: 22693548
- Kolaczinski JH, Reithinger R, Worku DT, Ocheng A, Kasimiro J, Kabatereine N, et al. Risk factors of visceral leishmaniasis in East Africa: a case-control study in Pokot territory of Kenya and Uganda. Int J Epidemiol. 2008; 37(2):344–52. https://doi.org/10.1093/ije/dym275 PMID: 18184669
- Ritmeijer K, Davidson R. Médecins Sans Frontières interventions against kala-azar in the Sudan, 1989–2003. Trans R Soc Trop Med Hyg. 2003; 97(6):609–13. PMID: 16134257
- Musa A, Khalil E, Hailu A, Olobo J, Balasegaram M, Omollo R, et al. Sodium stibogluconate (SSG) & paromomycin combination compared to SSG for visceral leishmaniasis in East Africa: a randomised controlled trial. PLoS Negl Trop Dis. 2012; 6(6):e1674. https://doi.org/10.1371/journal.pntd.0001674 PMID: 22724029
- Melaku Y, Collin SM, Keus K, Gatluak F, Ritmeijer K, Davidson RN. Treatment of kala-azar in southern Sudan using a 17-day regimen of sodium stibogluconate combined with paromomycin: a retrospective comparison with 30-day sodium stibogluconate monotherapy. Am J Trop Med Hyg. 2007; 77(1):89–94. PMID: 17620635
- Atia AM, Mumina A, Tayler-Smith K, Boulle P, Alcoba G, Elhag MS, et al. Sodium stibogluconate and paromomycin for treating visceral leishmaniasis under routine conditions in eastern Sudan. Trop Med Int Health. 2015; 20(12):1674–84. https://doi.org/10.1111/tmi.12603 PMID: 26427033
- Chappuis F, Alirol E, Worku DT, Mueller Y, Ritmeijer K. High mortality among older patients treated with pentavalent antimonials for visceral leishmaniasis in East Africa and rationale for switch to liposomal amphotericin B. Antimicrob Agents Chemother. 2011; 55(1):455–6. <u>https://doi.org/10.1128/AAC.</u> 01298-10 PMID: 21078947
- Ritmeijer K, Veeken H, Melaku Y, Leal G, Amsalu R, Seaman J, et al. Ethiopian visceral leishmaniasis: generic and proprietary sodium stibogluconate are equivalent; HIV co-infected patients have a poor outcome. Trans R Soc Trop Med Hyg. 2001; 95(6):668–72. PMID: <u>11816442</u>

- Croft S, Olliaro P. Leishmaniasis chemotherapy—challenges and opportunities. Clin Microbiol Infect. 2011; 17(10):1478–83. https://doi.org/10.1111/j.1469-0691.2011.03630.x PMID: 21933306
- Moore E, Lockwood D. Treatment of visceral leishmaniasis. J Glob Infect Dis. 2010; 2(2):151. https:// doi.org/10.4103/0974-777X.62883 PMID: 20606971
- 13. World Health Organisation. Report of a WHO informal consultation on liposomal amphotericin B in the treatment of visceral leishmaniasis. Rome: 16-04-2005.
- Balasegaram M, Ritmeijer K, Lima MA, Burza S, Ortiz Genovese G, Milani B, et al. Liposomal amphotericin B as a treatment for human leishmaniasis. Expert Opin Emerg Drugs. 2012; 17(4):493–510. https://doi.org/10.1517/14728214.2012.748036 PMID: 23167833
- Collin S, Davidson R, Ritmeijer K, Keus K, Melaku Y, Kipngetich S, et al. Conflict and kala-azar: determinants of adverse outcomes of kala-azar among patients in southern Sudan. Clin Infect Dis. 2004; 38 (5):612–9. https://doi.org/10.1086/381203 PMID: 14986243
- Seaman J, Mercer AJ, Sondorp HE, Herwaldt BL, Epidemic visceral leishmaniasis in southern Sudan: treatment of severely debilitated patients under wartime conditions and with limited resources. Ann Intern Med. 1996; 124(7):664–72. PMID: 8607595
- Horst ter R, Tefera T, Assefa G, Ebrahim AZ, Davidson RN, Ritmeijer K. Field evaluation of rK39 test and direct agglutination test for diagnosis of visceral leishmaniasis in a population with high prevalence of human immunodeficiency virus in Ethiopia. Am J Trop Med Hyg. 2009; 80(6):929–34. PMID: 19478251
- Barquet N, Domingo P, Caylà JA, González J, Rodrigo C, Fernández-Viladrich P, et al. Prognostic factors in meningococcal disease: development of a bedside predictive model and scoring system. JAMA. 1997; 278(6):491–6. PMID: 9256224
- Seymour DG, Green M, Vaz FG. Making better decisions: construction of clinical scoring systems by the Spiegelhalter-Knill-Jones approach. BMJ. 1990; 300(6719):223–6. <u>https://doi.org/10.1136/bmj.300.</u> 6719.223 PMID: 2106929
- 20. Katz M. Multivariable analysis: a primer for readers of medical research. Ann Int Med. 2003; 138 (8):644–50. PMID: 12693887
- Berkley J, Ross A, Mwangi I, Osier F, Mohammed M, Shebbe M, et al. Prognostic indicators of early and late death in children admitted to district hospital in Kenya: cohort study. BMJ. 2003; 326 (7385):361. https://doi.org/10.1136/bmj.326.7385.361 PMID: 12586667
- Coura-Vital W, Araujo VE, Reis IA, Amancio FF, Reis AB, Carneiro M. Prognostic factors and scoring system for death from visceral leishmaniasis: an historical cohort study in Brazil. PLoS Negl Trop Dis. 2014; 8(12):e3374. doi: 10.1371/journal.pntd.0003374. PMID: 25503575; PubMed Central PMCID: PMCPMC4263605.
- Cook NR. Use and misuse of the receiver operating characteristic curve in risk prediction. Circulation. 2007; 115(7):928–35. doi: 10.1161/CIRCULATIONAHA.106.672402. PMID: 17309939.
- Harrell FE, Lee KL, Mark DB. Multivariable Prognostic Models: Issues in Developing Models, Evaluating Assumptions and Adequacy, and Measuring and Reducing Errors1996. 361–87 p.
- Steyerberg EW, Vickers AJ, Cook NR, Gerds T, Gonen M, Obuchowski N, et al. Assessing the performance of prediction models: a framework for traditional and novel measures. Epidemiology. 2010; 21 (1):128–38. doi: 10.1097/EDE.0b013e3181c30fb2. PMID: 20010215; PubMed Central PMCID: PMCPMC3575184.
- Royston P, Moons K, Altman D, Vergouwe Y. Prognosis and prognostic research: developing a prognostic model. BMJ. 2009; 338:b604. https://doi.org/10.1136/bmj.b604 PMID: 19336487
- Altman DG, Vergouwe Y, Royston P, Moons KG. Prognosis and prognostic research: validating a prognostic model. BMJ. 2009; 338:b605. doi: 10.1136/bmj.b605. PMID: 19477892.
- Seaman J, Boer C, Wilkinson R, de Jong J, de Wilde E, Sondorp E, et al. Liposomal amphotericin B (AmBisome) in the treatment of complicated kala-azar under field conditions. Clin Inf Dis. 1995; 21 (1):188–93.
- Laman M, Aipit S, Bona C, Siba PM, Robinson LJ, Manning L, et al. Ultrasonographic assessment of splenic volume at presentation and after anti-malarial therapy in children with malarial anaemia. Malar J. 2015; 14(1):1.
- Abubakar A, Ruiz-Postigo JA, Pita J, Lado M, Ben-Ismail R, Argaw D, et al. Visceral leishmaniasis outbreak in South Sudan 2009–2012: epidemiological assessment and impact of a multisectoral response. PLoS Negl Trop Dis. 2014; 8(3):e2720. doi: 10.1371/journal.pntd.0002720. PMID: 24675573; PubMed Central PMCID: PMCPMC3967947.
- Collin SM, Coleman PG, Ritmeijer K, Davidson RN. Unseen Kala-azar deaths in south Sudan (1999– 2002). Trop Med Inter Health. 2006; 11(4):509–12.